Author's response to reviews

Title:A Her2-let-7-beta2-AR circuit affects prognosis in patients with Her2-positive breast cancer

Authors:

Dan Liu (liudan_bd@sina.com)
Limin Sun (wuying1984@163.com)
Tao Wang (wangtao6@hotmail.com)
Zhengyan Yang (zyy1984102@163.com)
Hongyu Chen (chy_0528@aliyun.com)
Liang Guo (gllsunmo369@sina.com)
Que Deng (dengque3s@163.com)
Yanjun Liu (425464861@qq.com)
Yuanfang Ma (mayf@henu.edu.cn)
Ning Guo (ning_guo@sina.com)
Ming Shi (sm200@sohu.com)

Version:3 Date:7 August 2015

Author's response to reviews:

Dear Editor,

Thank you very much for your letter. We appreciated the reviewer's comments very much. According to the kind suggestion of the reviewers and editor, we have performed extensive tests to validate our previous data and to answer the questions raised by the reviewers. We are hereby again submitting our revised manuscript for your consideration.

During the past several months, Dr. Que Deng (originally the seventh) performed a lot of experiments for the revised manuscript and he also worked very hard in the previous study. In the revised manuscript he is listed as one of the co-first authors.

We thank you again for your patience and encouragement during the process.

Best wishes,

Ming Shi and Ning Guo